Author's response to reviews

Title: The relationship between job satisfaction, burnout, and turnover intention among physicians from urban state-owned medical institutions in Hubei, China: a cross-sectional study

Authors:

Yimin Zhang (zym_03730@sina.com)
Xueshan Feng (xsfeng@fudan.edu.cn)

Version: 5 Date: 21 September 2011

Author's response to reviews:

1#The abstract should be identical between in the manuscript and the submission system.
2#all highlighting and tracking had been removed from the manuscript.
3#’Acknowledgements and Funding’ had been renamed to ’Acknowledgements’.
4#Though some tables had been re-laid to meet the formatting standard, I am not sure whether these requested problems had been well resolved.
5#Any typographical errors in this manuscript had been checked and revised.